\documentclass[twoside]{article}
\usepackage{ctex}
\usepackage{geometry}
\geometry{paperwidth=21cm,paperheight=29cm,top=1.5cm, bottom=1.5cm, left=2.54cm,right=2.54cm}
\usepackage{xcolor}
\usepackage[CJKbookmarks,colorlinks,bookmarksnumbered=true,pdfstartview=FitH,linkcolor=blue]{hyperref}
\usepackage[backend=biber,style=gb7714-2015,gbpub=true]{biblatex}%align=gb7714-2015
%\addbibresource[location=local]{egtest.bib}
\addbibresource[location=local]{example.bib}
\addbibresource[location=local]{egspecialchar.bib}
%\renewcommand{\bibfont}{\zihao{5}}
%\setlength{\bibitemsep}{0ex}
%\setlength{\bibnamesep}{0ex}
%\setlength{\bibinitsep}{0ex}
%\renewcommand{\thefootnote}{\textcircled{\tiny\arabic{footnote}}}

\begin{document}

%%%测试作者年/biblatex版本兼容性的文件
%%%test file for author year style

\section{文献表}
\begin{enumerate}
  \item 学位论文引用\cite{CALMS1965--,马欢2011-27-27,吴云芳2003--,张若凌2004--,张志祥1998--}
\end{enumerate}

{
%\hyphenation{kurose-gawa}
%\hyphenpenalty=1000
%\tolerance=500
\printbibliography%[heading=subbibintoc,title=【参考文献】]
}
\end{document} 